

ARCHIVIO ISTITUZIONALE DELLA RICERCA

Alma Mater Studiorum Università di Bologna Archivio istituzionale della ricerca

Time spent on distance learning moderates changes in teachers' work-related well-being one year after the first school closures

This is the final peer-reviewed author's accepted manuscript (postprint) of the following publication:

Published Version:

Soncini, A., Politi, E., Matteucci, M.C. (2024). Time spent on distance learning moderates changes in teachers' work-related well-being one year after the first school closures. SCHOOL PSYCHOLOGY, 39(6), 711-718 [10.1037/spq0000575].

Availability:

This version is available at: https://hdl.handle.net/11585/969510 since: 2024-05-14

Published:

DOI: http://doi.org/10.1037/spq0000575

Terms of use:

Some rights reserved. The terms and conditions for the reuse of this version of the manuscript are specified in the publishing policy. For all terms of use and more information see the publisher's website.

This item was downloaded from IRIS Università di Bologna (https://cris.unibo.it/). When citing, please refer to the published version.

(Article begins on next page)

Time Spent on Distance Learning Moderates Changes in Teachers' Work-Related Well-Being

One Year After the First School Closures

Soncini A., Politi E. & Matteucci M. C.

Soncini, A., Politi, E., & Matteucci, M. C. (2023). Time spent on distance learning moderates changes in teachers' work-related well-being one year after the first school closures. *School Psychology*. Advance online publication. <u>https://doi.org/10.1037/spq0000575</u>

Abstract

It is now well-documented that school closures enforced at the beginning of the COVID-19 pandemic impaired teachers' well-being. Yet, only a few studies tracked changes in teachers' wellbeing during the subsequent phases of the pandemic; phases that were characterised by the discontinuous implementation of in-person teaching and distance learning. To fill this gap, we conducted a follow-up study at the end of the school year 2020-2021 (May – June 2021, T2), administering an online questionnaire to Italian teachers (N = 240) who had previously taken part in a data collection conducted at the end of the first school closures (May – June 2020, T1). Our first aim was to monitor changes in teachers' psychological and work-related well-being between T1 and T2. Our second aim was to assess whether time spent on distance learning moderates these changes in psychological and work-related well-being. Results showed that teachers' psychological wellbeing decreased between T1 and T2, whereas work-related well-being increased. What is more, time spent on distance learning moderated the general increase in work-related well-being observed at T2: the longer teachers implemented distance learning during the school year 2021, the less their work-related well-being increased. In conclusion, although it seems that teachers have adapted to the changes associated with the first school closures, this study showed that distance learning remains a possible risk factor for teachers' well-being.

Keywords: Teachers, distance learning, psychological well-being, work-related well-being, school closures

Public Significance Statement

This follow-up study explored changes in teachers' well-being between the first school closures and the end of the following school year in Italy. While teachers' psychological well-being decreased, results show that their work-related well-being increased over time. However, the longer teachers implemented distance learning, the weaker their increase in work-related well-being. Distance learning remains a stress factor for teachers that risks compromising their work-related well-being.

Time Spent on Distance Learning Moderates Changes in Teachers' Work-Related Well-Being One Year After the First School Closures

In 2020, more than 63 million teachers worldwide were affected by school closures enforced to stop the COVID-19 pandemic. Most of them lacked the necessary support and training to switch to distance learning (UNESCO, 2020). Numerous empirical investigations have demonstrated that this abrupt shift has led to a decline in teachers' well-being. (e.g., Pressley et al., 2021; Hilger et al., 2021). Some countries had to keep schools closed or only partially open during the school year after the first school closures due to the pandemic waves that followed the first one in March 2020, irregularly combining in-person teaching and distance learning (UNESCO, 2021). Despite the potential negative impact of these measures on the already compromised teachers' well-being (e.g., Pressley et al., 2021), only a few studies monitored possible changes in teachers' conditions across time (e.g., Nabe-Nielsen et al., 2021; Kim et al., 2022). To complement our findings on Italian teachers' well-being at the end of the first school closures (2020), we conducted a follow up study to document longitudinal changes in teacher well-being one year later. In addition, we analysed the effect of time spent on distance learning, implemented discontinuously during the school year 2020-2021, on changes in teachers' well-being.

Teachers' Well-Being During the COVID-19 Pandemic

Teaching is considered one of the most stressful jobs (Johnson et al., 2005), and teachers suffer more from mental and psychosomatic disorders than other occupational groups (e.g., Redin & Erro-Garces, 2020; Viac & Fraser, 2020). The school closures imposed worldwide in March 2020 to stop the COVID-19 pandemic consistently affected teachers, decreasing their well-being. Indeed, teachers had to adjust to a completely new working environment and learn to implement online teaching activities. This unexpected change led to a general decrease in their well-being as evidenced by many studies conducted in different countries (e.g., Soncini et al., 2021; Pressley et al., 2021; Hilger et al., 2021; Zhen et al., 2021; Silva et al., 2021).

Recent studies have shown that teachers reported low levels of well-being even during the school year that followed the first school closures, both when schools were completely closed (Lizana & Lera, 2022; Jakubowski & Sitko-Dominik, 2021) and when schools were fully or partially reopened (Ozamiz-Extrebarria et al., 2021). To the best of our knowledge, the only two available studies that monitored teachers' well-being across time through a longitudinal approach showed a general decrease in well-being from the first school closures and throughout the school year after, when teachers resumed in-person teaching (Nabe-Nielsen et al., 2021; Kim et al., 2022). However, in these two studies, well-being was broadly defined in terms of general mental health and the authors did not include more nuanced measures.

To enrich extant research and tackle different facets of teachers' well-being, in this study, we focused on multiple aspects of well-being. Following the model proposed by Diener (2009) we considered emotional well-being (i.e., positive moods) and life satisfaction (i.e., cognitive appraisal of life as a whole) as two aspects of *psychological well-being*. In addition, since the challenges faced by teachers during the pandemic mainly concerned their job, we also assessed two complementary aspects of *work-related well-being* (Rothmann, 2008), namely work engagement, (i.e., a job-related persistent positive state of mind; Schaufeli, 2006), and emotional exhaustion (i.e., the main symptom of the burnout syndrome characterised by extreme fatigue and energy depletion; Maslach & Jackson, 1981). For clarity, throughout the manuscript, we will refer to psychological well-being to indicate emotional well-being and life satisfaction and work-related well-being to indicate emotional exhaustion and work engagement.

Distance Learning: A Major Stressor for Teachers During COVID-19 Pandemic

When the first school closures were enforced in several countries in March 2020 (UNESCO, 2021), distance learning became the only way to provide pupils with education and instruction. However, working from home increased teachers' workload, uncertainty, and work-related threats (Nabe-Nielsen et al., 2021; Silva et al., 2021). In this regard, teachers expressed a general preference for in-person teaching over distance learning, which was considered difficult to manage (e.g., Mseleku, 2020). Based on a large number of teachers–a subsample of whom also participated in this follow-up study–Soncini and colleagues (2021) found that distance learning was perceived as one of the main stressors for teachers during the first school closures.

After the first school closures, some countries kept schools closed or partially closed, implementing distance learning and in-person teaching discontinuously. In Italy, in-person teaching was officially resumed in September 2020 (at the start of the school year 2020-2021). However, school closures were enforced several times from November 2020 to March 2021, either throughout the country or in particularly high-risk areas, and depending on the school level: Primary and middle schools could remain open in high-risk areas, whereas secondary schools had to switch to distance learning (Decree of the Prime Minister – November 2020). In addition, whenever a case of COVID-19 was diagnosed in a classroom, the whole class had to mandatorily switch to distance learning to keep schools open while limiting the risk of contagion (Decree of the Prime Minister – September 2020). As a result, Italian school closures were very fragmented compared to other European countries (e.g., Hale et al., 2021), and the amount of time teachers spent on distance learning varied considerably (Bovini & De Filippis, 2021). Most likely, this fragmented situation exacerbated the difficulties related to the implementation of distance learning that Italian teachers had already reported during the first school closures (Soncini et al., 2021; Toto et al., 2021; Truzoli et al., 2020).

The Present Study

Although evidence of a decrement in the well-being of teachers during the first school closures has been provided, only a few studies tracked teachers' well-being during the year following the initial school closures. Furthermore, no study has yet explored the effects of time spent on distance learning on possible changes in teachers' well-being registered during the school year that followed the first school closures. Understanding these phenomena is particularly important for countries where distance learning has been applied discontinuously, such as Italy.

Accordingly, this study's aim was twofold. First, to compare the psychological (i.e., emotional wellbeing and life satisfaction) and work-related well-being (i.e., work-engagement and emotional exhaustion) of teachers assessed during the last two months of the school year influenced by the first school closures (May – June 2020, T1) with psychological and work-related well-being assessed the last two months of the following school year (May – June 2021, T2). Second to assess whether the time spent on distance learning impacted on the change in teachers' psychological wellbeing and work-related well-being between T1 and T2. In line with these aims, we formulated the following hypotheses:

H1: We expected a general decrease in teachers' psychological and work-related well-being between T1 and T2.

H2: We expected that time spent on distance learning moderated the general decrease in teachers' psychological and work-related well-being over time, so that teachers who implemented distance learning more often during the school year 2020-2021 reported a more severe deterioration in psychological and work-related well-being, compared to teachers who implemented distance learning less often.

The present study is the follow-up phase of a large study conducted in Italy at the end of the year when school closures were first enforced (March 2020 - 2020). This prior cross-sectional study was set to assess teachers' psychological and work-related well-being during the last two months of school (May – June 2020) to understand the impact of this challenging period on Italian teachers. The data collection of the prior study was carried out by sending e-mails to all the primary, middle, and secondary school principals of three Italian regions (Emilia-Romagna, Northern Italy; Marche, Central Italy and Sardinia, Southern Italy), asking them to share the link to an online questionnaire among teachers working at their schools. A total of 1036 teachers took part in the first data collection and they were asked to leave their e-mail addresses to be contacted again the following year to participate in the follow-up. Results based on the first data collection can be found in Soncini and colleagues (2021; 2022). The present study reports the results of the follow-up study set

7

out during the last two months of the school year that followed the first Italian school closures (May – June 2021).

Method

Participants

This study was based on a sub-sample of teachers who had already participated in a large cross-sectional study conducted during the first school closures (Soncini et al., 2021; 2022). Protocols and materials were approved by the third author's University Ethical Board (protocol 2870, date 29/05/2020). A total sample of 1036 primary, middle, and secondary school teachers ($M_{age} = 49.67$, SD = 9.28, range 21-67, 85.91% female; 34.72% primary school teachers, 16.21 middle school teachers, and 49.07% secondary school teachers) took part in the first data collection (May – June 2020, T1). Among the total sample, 424 teachers (41.05% of 1036; $M_{age} = 49.19$, SD = 9.41, range 27-67, 84.5% female) left their e-mail address to be contacted again to take part in the follow-up study, which was conducted between May – June 2021 (T2). The final sample was composed of 240 teachers (23.2% of 1036, 56.6% of 440; $M_{age} = 49.61$, SD = 9.17, range 27-66, 82.9% female), who agreed to complete the follow-up questionnaire.

To check for possible sampling biases between teachers who only participated in the first data collection and those who participated in both data collections, we compared their responses to socio-demographic variables (age, gender, school level, teaching experience) and psychometric scales (Table S1 and S2, Supplementary Online Materials). None of the socio-demographic variables differed significantly, apart from school level, X^2 (2) = 6.66, p = .036. Indeed, fewer middle school teachers participated in the follow-up study (n = 26 out of 240; 10.80%) compared to those who participated only in the first data collection (n = 142 out of 796; 17.80%, see Table S2). Since few middle school teachers participated in the follow-up study, and the Italian regulations on distance learning at primary and middle schools differed from those at secondary school level, we created a new variable that aggregated primary and middle school teachers (n = 115, 47.92%) and differentiated them from secondary school teachers (n = 125, 52.13%).

Measures

The following psychometric scales were administered at both T1 and T2.

Emotional Well-being

The World Health Organization Well-being Index (see Topp et al., 2015) was used to assess emotional well-being. This scale measures the frequency of positive affective states and consists of 5 items (e.g., *During the last two weeks, I felt happy and in a good mood*) and a 6-point Likert scale (0 = ``Never'', 5 = ``Always''). The internal consistency was good $(T1\omega = .80, T2\omega = .91)$.

Life Satisfaction

The Italian version (Di Fabio & Ghizzani, 2006) of the Satisfaction with Life Scale (Diener et al., 1985) was used to assess life satisfaction. This scale consists of 5 items (e.g., *In many ways, my life is close to my ideal.*) assessed through a 7-point Likert scale (1 = "Strongly disagree", 7 = "Strongly agree"). The internal consistency was good ($T1\omega = .89, T2\omega = .92$).

Emotional Exhaustion

The Italian version (Sirigatti et al., 1988) of the Maslach Burnout Inventory- MBI (Maslach & Jackson, 1981) was used to measure teachers' emotional exhaustion. This scale consists of 5 items (e.g., "*I feel exhausted at the end of the working day*"), assessed via a 7- points Likert scale (0 = "Never", 6 = "Every day"). The internal consistency was good ($T1\omega = .88$, $T2\omega = .93$).

Work Engagement

The Italian version (Balducci et al., 2010) of the Utrecht Work Engagement Scale (UWES-9; Schaufeli et al., 2006) was used to assess teachers' work engagement. The scale consists of 9 items measured using a 7-point Likert scale (0 = "Never", 6 = "Every day"), and divided into 3 subdimensions: Vigor (e.g., *I feel full of energy in my work*), Dedication (e.g., *I am proud of my work*), Absorption (e.g., *I am immersed in my work*). The internal consistency of the aggregated scale was good ($T1\omega = .90$, $T2\omega = .91$).

Time Spent on Distance Learning

This variable was only included in the follow-up questionnaire. It measured how much time teachers spent on distance learning during the school year 2020–2021 ("How much time in the school year 2020–2021 did you spend on distance learning? Please make an estimate considering the entire school year from September to June"). The scale ranged from 0 (distance learning implemented for less than one month) to 9 months (distance learning implemented for almost the entire school year)¹.

Data Analysis

Analyses were conducted using SPSS 28.² First, we tested the degree of overlap between time spent on distance learning (i.e., moderator) and school level (i.e., covariate), to detect possible problems of multicollinearity between these two variables. Second, we checked for missing value patterns and imputed data that were not missing completely at random. We concluded this section on preliminary analysis, testing whether the four dependent measures of teachers' psychological well-being (i.e., emotional wellbeing and life satisfaction) and work-related well-being (i.e., emotional exhaustion and work engagement) followed a normal distribution.

Once these assumptions were verified, we tested the hypotheses. According to H1, we expected a general decrease in teachers' psychological and work-related well-being between T1 and T2. To test H1, we assessed whether scores of teachers' psychological and work-related well-being between T1 and T2 differed significantly from zero, using a series of paired-sample t-tests.

According to H2, we expected that time spent on distance learning moderated the general decrease in teachers' psychological and work-related well-being. To test H2, we ran a series of moderation analyses in a two-instance repeated measures design, using the SPSS macro MEMORE

¹ Example: If a teacher applied distance learning during the national school closure between November and January (3 months), adding a further three weeks between February and March because their region was a high-risk region and a further two weeks in May because several cases were recorded in their classes, they should have responded 5 months in total.

² The complete dataset and the syntax are available on Open Science Framework <u>https://osf.io/vfyba/?view_only=6da41358112b4013af25f7d9dd636384</u>

v. 2.1.2 (Montoya, 2019). Changes in teachers' psychological and work-related well-being between T1 and T2 were modelled as a within-participant repeated measures factor. Time spent on distance learning was modelled as a between-participant fixed factor. If a significant interaction between the within and between factors was found, the interaction was probed using the Johnson–Neyman procedure. Compared to standard probing techniques—such as the decomposition of main effects at arbitrary moderator cut-off points set at ± 1 SD—the Johnson-Neyman procedure decomposes the interaction effect across the entire distribution range of the moderator variable, thus resulting in more informative and precise (Hayes, 2022; Johnson & Neyman, 1936). By using the Johnson-Neyman procedure, indeed, we were able to estimate the exact amount of time spent on distance learning at which the changes in teachers' psychological and work-related well-being became significant.

A major advantage of repeated measures is that each participant serves "as their own statistical control" (Montoya, 2019). The noise due to unmeasured confounding factors (e.g., stable dispositional differences, individual professional experiences, personal background) is thus considerably reduced in within-subject designs (Judd et al., 1996). Unless the within-effect of covariates is expected to vary between time points, the estimation of control variables (in this case, gender, age, and years of experience) in the model is not necessary. To decide whether to include the school level as a covariate we estimated the degree of overlap with time spent in distance learning and their multicollinearity. Indeed, during the school year that followed the first school closures, schools closed differently depending on the school level: Most of the primary and middle schools remained open throughout the school year, while secondary schools in high-risk areas switched to distance learning.

Results

Preliminary Results

First, we checked the degree of overlap between time spent on distance learning and school level. As shown in Figure 1, the distribution of time spent on distance learning between primary and middle school teachers, on the one hand, and secondary school teachers, on the other, immediately suggests a sizable difference between the two groups. An independent samples t-test confirmed a large difference in time spent on distance learning, t(238) = -18.76, p < .001, d = -2.42, reported by primary and middle school teachers (M = 1.28, SD = 0.81), as compared to secondary school teachers (M = 5.36, SD = 2.19). The strong interdependence between school level and time spent on distance learning resulted in low tolerance levels (tolerance = .338, VIF = 2.955). These low tolerance levels may indicate problems of multicollinearity (Alin, 2010), and they are likely to decrease the accuracy of estimated effects of the variable of interest on the dependent variable (York, 2018).

Figure 1

Frequencies of Months Spent on Distance Learning Reported Separately for Primary and Middle School Teachers, and for Secondary School Teachers



The inclusion of both variables in the model would be statistically problematic, and it would be conceptually artificial to test the contribution of one predictor while controlling for the other. Since our main hypotheses referred to time spent on distance learning and not to idiosyncratic differences due to variations in school level, we did not control for school level in the main moderation analyses. Further confirming the conceptual overlap between these two variables, additional analyses using school level instead of time spent on distance learning showed the same results (see Supplementary Online Materials).

Second, we handled missing values. The Little's MCAR test was significant, suggesting that data were not missing completely at random, $\chi^2(240) = 375.291$, p < .001. Among the variables, only time spent on distance learning presented 9.2% (n = 22) of missing data, and thus, we used Expectation Maximization algorithm in SPSS to perform a single data imputation for missing data. Although multiple imputation techniques generally outperform single imputation (e.g., Donders et al., 2006; Lin & Tsai, 2020), MEMORE does not include analytical procedure to pool multiple datasets.

Third, we tested normal distribution through levels of skewness and kurtosis of all psychometric scales, which are fully reported in Table S3 in the Supplementary Online Materials. Only work engagement resulted in a moderate violation of normality. As reported by Russell and Dean (2000), the bootstrap method represents the best way to solve the problem of violation of normality in complex moderation analyses (specifically in case of a two-instance repeated measures design where variable transformation may cause a miscalculation of difference scores). However, the macro MEMORE, used for the moderated analysis in this study, does not allow for bootstrapping. For this reason, we did not transform the variable and accepted possible estimation errors on work engagement.

Hypotheses Testing

Change in Teachers' Psychological and Work-Related Well-Being Between T1 and T2

13

Raw means and standard deviations of psychological well-being (i.e., emotional well-being and life satisfaction), and work-related well-being (i.e., emotional exhaustion and work engagement) are reported in Table 1. Results extracted from a series of paired-sample t-tests revealed significant time variations in both psychological and work-related well-being but in opposite directions. In line with H1, psychological well-being deteriorated over time: As compared to 2020, teachers reported lower emotional well-being and life satisfaction. Contrary to H1, however in 2021, work-related well-being improved over time: As compared to 2020, teachers reported lower emotional exhaustion and greater work engagement in 2021 (see Table 1).

Table 1

Changes in Teachers' Psychological and Work-related Well-being Between T1 and T2, Together with Observed Means and Standard Deviations

Psychometric scales		T1		T2				
	Range	М	SD	М	SD	t(239)	р	d
Psychological well-being								
Emotional well-being	0-5	3.05	0.97	2.43	1.15	-9.22	<.001	60
Life satisfaction	1 - 7	5.01	1.19	4.81	1.39	-3.16	.002	20
Work-related well-being								
Emotional Exhaustion	0-6	3.73	1.66	2.96	1.74	-7.89	<.001	51
Work Engagement	0-6	4.78	1.08	5.02	0.96	3.91	<.001	.25

Note: Results were extracted from a series of paired-sample t-tests for each psychometric scale separately. Effect sizes are expressed using Cohen d' and shown on the right-hand side of the table and represents the difference between T2 and T1.

Moderating Effects of Time Spent on Distance Learning on Changes in Teachers' Well-Being

In relation to psychological well-being, results extracted from MEMORE revealed that time spent on distance learning did not moderate the general deterioration observed between T1 and T2

of emotional well-being, b = -0.014 (*SE* = .025), p = .593, 95% CI [-0.079, 0.052], $R^2 = .001$, and life satisfaction, b = -0.001 (*SE* = .024), p = .679, 95% CI [-0.071, 0.051], $R^2 = .001$. Conversely as for work-related well-being, time spent on distance learning did moderate the observed decrease in emotional exhaustion, b = 0.131 (*SE* = .036), p < .001, 95% CI [0.038, 0.224], $R^2 = .053$, and the observed increase in work engagement to a lower extent, b = -0.047 (*SE* = .023), p = .043, 95% CI [-0.109, 0.013], $R^2 = .017$.

We investigated interaction effects using the Johnson-Neyman procedure (Johnson & Neyman, 1936). Transition points (JN₁) and significance regions for conditional changes in teachers' emotional exhaustion and work engagement are reported in Figure 2 and Figure 3, respectively. JN₁ are points in the distribution of the moderating variable where confidence intervals intersect zero on the y-axis. Significance regions are those portions of the distribution of the outcome variable that differ significantly between time points. Concerning emotional exhaustion, teachers (n = 197) who had spent less than 6.63 months on distance learning during the school year that followed the first school closures reported lower levels of emotional exhaustion at T2 as compared to T1. This threshold level includes all primary and middle school teachers, and 65.60% (n = 82) of secondary school teachers (Figure 1). As for work engagement, teachers (n = 167) who had spent 4.68 months or less on distance learning during the school year that followed the first school teachers (Figure 1). As for work engagement to T1. This threshold level includes all primary and middle school year that followed the first school teachers (Figure 1). As for work engagement, teachers (n = 167) who had spent 4.68 months or less on distance learning during the school year that followed the first school closures reported greater work engagement at T2 as compared to T1. This threshold level includes all primary and middle school teachers (Figure 1). Confirming H2, time spent on distance learning dampened the general improvement in work-related well-being observed over time.

Figure 2

Conditional Change in Teachers' Emotional Exhaustion as a Function of Time Spent on Distance



Learning

Figure 3

Conditional Change in Teachers' Work Engagement as a Function of Time Spent on Distance

Learning



Discussion

The aim of this follow-up study was twofold: To monitor whether teachers' psychological well-being (i.e., emotional well-being and life satisfaction) and work-related well-being (i.e., emotional exhaustion and work engagement) changed one year after the first school closures enforced to stop to the COVID-19 pandemic; and to estimate whether time spent on distance learning during the school year that followed the first school closures moderated these changes. We collected data during the last two months of the first school closures enforced in Italy (May – June 2020, T1) and during the last two months of the school year that followed (May – June 2021, T2) through an online questionnaire administered to Italian teachers both at T1 and at T2.

Concerning psychological well-being, results were in line with previous studies that reported a gradual deterioration of teachers' well-being across the pandemic weaves (Jakubowski & Sitko-Dominik, 2021; Lizana & Lera, 2022; Nabe-Nielsen et al., 2021). However, our results showed that teachers' work-related well-being increased over time: Compared to T1 (March – June 2020), at T2 (May – June 2021) levels of work engagement increased, and emotional exhaustion decreased. This result differs from findings from previous studies that reported low levels of teachers' work-related well-being even during the school year following the first school closures (Kotowsky et al., 2022; Klusmann et al., 2023; Raducu & Stanculesku, 2022). However, it is important to note that these studies were cross-sectional and did not measure a change between two different moments of the pandemic in the same teachers.

Because we measured multiple aspects related to both psychological and work-related wellbeing, this study contributed to the existing literature by showing that psychological and workrelated well-being evolved differently during the school year that followed the first school closures. Indeed, previous studies focused only on aspects of psychological well-being (e.g., depression or anxiety, Lizana & Lera, 2022), or work-related well-being (e.g., Klusmann et al., 2023) without analysing them together. According to our results, besides the psychological fatigue cumulated after one year of the pandemic, teachers adjusted positively to new working conditions and gained a new sense of engagement in their job.

Nuancing these promising findings, and partially in line with our second hypothesis, the general improvement in work-related well-being was moderated by the amount of time spent on distance learning during the school year 2020-2021. Accordingly, the longer teachers implemented distance learning, the less their work-related well-being increased. This moderating effect of time spent on distance learning was particularly strong in relation to teachers' emotional exhaustion. In line with other findings collected during the first phases of the pandemic (e.g., Soncini et al., 2021; Mseleku, 2020), distance learning remained a major stressor that led to negative consequences for teachers' emotional adjustment. This result is in line with stress-related theories (i.e., Lazarus & Folkam, 1984) which posit that the impact of potential stressors on well-being is determined by people's perceptions. Most likely, distance learning was experienced as a challenging aspect of the jobwork by the teachers even during the school year that followed the first school closures. Consequently, spending more time on distance learning became a challenge to be faced rather than an opportunity to be seized.

This study has some limitations that should be noted. First, although teachers included in the two subsamples did not differ on most of the key socio-demogrphic variables, attrition in the follow-up study was quite high (only 23.10% of the total sample participated in T2), reducing therepresentativeness of our sample. In addition, the results may have been influenced by a self-selection bias (Heckman, 2010) that caused only the most distressed teachers to participate in the follow-up study. Second, the scale used to measure time spent on distance learning was based on a subjective evaluation given by teachers, rather than the actual time spent on such activity as reported on class registers. This self-reported scale brings endogeneity in the model because it relies on teachers' perceptions and memories, which may differ according to their level of psychological and work-related well-being. In addition, given the Italian regulations on distance learning, school level was closely overlapping with the time spent on distance learning, making it impossible to

18

properly disentangle the two factors and their unique relation with work-related and psychological well-being. Third, work engagement at T2 was not normally distributed and this may have reduced the accuracy of effects estimation. Finally, we did not measure other possible moderating factors, such as social or technical support received from schools (e.g., Soncini et al., 2022). Overall, future investigations should tease apart the potential benefits of hybrid forms of distance learning and identify conditions under which distance learning is viewed as an opportunity by teachers, as well as propose a detailed investigation which encompasses more variables.

Despite these limitations, this follow-up study highlighted the need to disentangle the psychological and work-related well-being of teachers and to monitor both aspects at regular intervals. In addition, it pointed out that distance learning represented a major stressor for teachers even after the first school closures. Taken together, our findings may help the development of tailored programs aimed at supporting teachers' well-being, a key factor to recover educational losses and manage long-term teaching transformations caused by the COVID-19 pandemic (UNESCO, 2021).

References

- Alin, A. (2010). Multicollinearity. Wiley Interdisciplinary Reviews: Computational Statistics, 2(3), 370-374. Retrieved from: 10.1002/wics.84
- Balducci, C., Fraccaroli, F., & Schaufeli, W. B. (2010). Psychometric properties of the Italian version of the Utrecht Work Engagement Scale (UWES-9). *European Journal of Psychological Assessment*, 26, 143-149. <u>https://doi.org/10.1027/1015-5759/a000020</u>
- Bovini, G., & De Philippis, M. (2021). Evidence on how distance learning is conducted and effects for Italian families. *Italian Bank Report*. Retrieved from: <u>https://www.bancaditalia.it/pubblicazioni/note-covid</u>
- Decree of the Prime Minister September 2020 "Urgent provisions to support needs for the start of the school year related to the epidemiological emergency from COVID-19". Retrieved from: https://www.gazzettaufficiale.it/eli/id/2020/09/08/20G00134/sg
- Decree of the Prime Minister November 2020 "Additional urgent measures to cope with the epidemiological emergency from COVID-19". Retrievede from: https://www.trovanorme.salute.gov.it/norme/dettaglioAtto?id=76993
- Diener, E. (2009). Subjective well-being. In E. Diener (Ed.), The Science of Well-Being: The Collected Works of Ed Diener. (pp. 11-58) Springer Science and Business Media. http://doi.org/10.1007/978-90-481-2350-6_2
- Diener, E., Emmons, R. A., Larsen, R. J., & Griffin, S. (1985). The satisfaction with life scale. *Journal of Personality Assessment*, 49(1), 71-75. Retrieved from. https://emmons.faculty.ucdavis.edu/wp-content/uploads/sites/90/2015/08/1985_5-SWLS.pdf
- Di Fabio, A., & Ghizzani, F. (2006). Satisfaction With Life Scale (SWLF) versione italiana. *Counseling*, 2, 201-211. http://hdl.handle.net/2158/373633
- Donders, A. R. T., Van Der Heijden, G. J., Stijnen, T., & Moons, K. G. (2006). A gentle introduction to imputation of missing values. *Journal of Clinical Epidemiology*, 59(10), 1087-1091. https://doi.org/10.1016/j.jclinepi.2006.01.014

- Hale, T., Angrist, N., Goldszmidt, R., Kira, B., Petherick, A., Phillips, T., Webster, S., Cameron-Blake, E., Hallas, L., Majumdar, S., & Tatlow, H. (2021). A global panel database of pandemic policies Oxford COVID-19 Government Response Tracker. *Nature Human Behaviour*, 5, 529–538. <u>https://doi.org/10.1038/s41562-021-01079-8</u>
- Hayes, A. F. (2022). Introduction to Mediation, Moderation, and Conditional Process Analysis Third Edition: A Regression-Based Approach. Guilford Press.
- Heckman, J. J. (2010). Selection bias and self-selection. In Durlauf S. N. & Blume L. E. (Eds), *Microeconometrics* (pp. 242-266). The New Palgrave Economics Collection (NPHE).
- Hilger, K. J., Scheibe, S., Frenzel, A. C., & Keller, M. M. (2021). Exceptional circumstances: Changes in teachers' work characteristics and well-being during COVID-19 lockdown. *School Psychology*, 36(6), 516- 532. https://doi.org/10.1037/spq0000457
- Jakubowski, T. D., & Sitko-Dominik, M. M. (2021). Teachers' mental health during the first two waves of the COVID-19 pandemic in Poland. *PloS One*, *16*(9), 1-25. <u>https://doi.org/10.1371/journal.pone.0257252</u>
- Johnson, P. O., & Neyman, J. (1936). Test of certain linear hypotheses and their applications to some educational problems. In J. Neyman & O. Christ (Eds.), *Statistical Research Memories* (pp. 57–93). Cambridge University Press.
- Johnson, S., Cooper, C., Cartwright, S., Donald, I., Taylor, P., & Millet, C. (2005). The experience of work-related stress across occupations. *Journal of Managerial Psychology*, 20(2), 178-187. <u>https://doi.org/10.1108/02683940510579803</u>
- Judd, C. M., McClelland, G. H., & Smith, E. R. (1996). Testing treatment by covariate interactions when treatment varies within subjects. *Psychological Methods*, 1, 366– 378. <u>https://doi.org/10.1037/1082-989X.1.4.366</u>
- Kim, L. E., Oxley, L., & Asbury, K. (2022). "My brain feels like a browser with 100 tabs open": A longitudinal study of teachers' mental health and well-being during the COVID-19

pandemic. *British Journal of Educational Psychology*, 92(1), 299-318. https://doi.org/10.1111/bjep.12450

- Klusmann, U., Aldrup, K., Roloff-Bruchmann, J., Carstensen, B., Wartenberg, G., Hansen, J., & Hanewinkel, R. (2023). Teachers' emotional exhaustion during the Covid-19 pandemic: Levels, changes, and relations to pandemic-specific demands. *Teaching and Teacher Education*, 121, 103908. <u>https://doi.org/10.1016/j.tate.2022.103908</u>
- Kotowski, S. E., Davis, K. G., & Barratt, C. L. (2022). Teachers Feeling the Burden of COVID-19: Impact on Well-being, Stress, and Burnout. *Work*, 71, 407 – 415. https://doi.org/2022 10.3233/WOR-210994
- Lazarus, R. S., & Folkman, S. (1984). Stress, appraisal, and coping. New York: Springer
- Lin, W. C., & Tsai, C. F. (2020). Missing value imputation: a review and analysis of the literature (2006–2017). *Artificial Intelligence Review*, *53*, 1487-1509. https://doi.org/10.1007/s10462-019-09709-4
- Lizana, P. A., & Lera, L. (2022). Depression, Anxiety, and Stress among Teachers during the Second COVID-19 Wave. International Journal of Environmental Research and Public Health, 19(10), 59-68. <u>https://doi.org/10.3390/ijerph19105968</u>
- Mansfield, E. R., & Helms, B. P. (1982). Detecting multicollinearity. *The American Statistician*, *36*(3a), 158-160. <u>https://doi.org/10.1080/00031305.1982.10482818</u>
- Maslach, C., & Jackson, S. E. (1981). The measurement of experienced burnout. *Journal of Organizational Behavior*, 2, 99-113. <u>https://doi.org/10.1002/job.4030020205</u>
- Montoya, A. K. (2019). Moderation analysis in two-instance repeated measures designs: Probing methods and multiple moderator models. *Behavior Research Methods*, 51(1), 61–82. https://doi.org/10.3758/s13428-018-1088-6

- Mseleku, Z. (2020). A literature review of E-learning and E-teaching in the era of Covid-19 pandemic.
 International Journal of Innovative Science and Research Technology, 57(52), 588-597.
 Retrieved from: <u>https://ijisrt.com/assets/upload/files/IJISRT20OCT430.pdf</u>
- Nabe-Nielsen, K., Christensen, K. B., Fuglsang, N. V., Larsen, I., & Nilsson, C. J. (2022). The effect of COVID-19 on schoolteachers' emotional reactions and mental health: Longitudinal results from the CLASS study. *International Archives of Occupational and Environmental Health*, 95(4), 855-865. <u>https://doi.org/10.1007/s00420-021-01806-8</u>
- Ozamiz-Etxebarria, N., Berasategi Santxo, N., Idoiaga Mondragon, N., & Dosil Santamaría, M. (2021). The psychological state of teachers during the COVID-19 crisis: The challenge of returning to face-to-face teaching. *Frontiers in Psychology*, 11, 1-10. https://doi.org/10.3389/fpsyg.2020.620718
- Pressley, T., Ha, C., & Learn, E. (2021). Teacher stress and anxiety during COVID-19: An empirical study. *School Psychology*, *36*(5), 367-376. <u>https://doi.org/10.1037/spq0000468</u>
- Răducu, C. M., & Stănculescu, E. (2022). Personality and socio-demographic variables in teacher burnout during the COVID-19 pandemic: a latent profile analysis. *Scientific Report*, 12, 14272. <u>https://doi.org/10.1038/s41598-022-18581-2</u>
- Redín, C. I., & Erro-Garcés, A. (2020). Stress in teaching professionals across Europe. *International Journal of Educational Research*, *103*, 101623. <u>https://doi.org/10.1016/j.ijer.2020.101623</u>
- Rothmann, S. (2008). Job satisfaction, occupational stress, burnout and work engagement as components of work-related wellbeing. *SA journal of Industrial Psychology*, *34*(3), 11-16. Retrived from: http://www.scielo.org.za/pdf/sajip/v34n3/02.pdf
- Schaufeli, W. B., Bakker, A. B., & Salanova, M. (2006). The measurement of work engagement with a short questionnaire: A cross-national study. *Educational and Psychological Measurement*, 66(4), 701-716. https://doi.org/10.1177/0013164405282471
- Silva, D. F. O., Cobucci, R. N., Lima, S. C. V. C., & de Andrade, F. B. (2021). Prevalence of anxiety, depression, and stress among teachers during the COVID-19 pandemic: A PRISMA-

compliant systematic review. *Medicine*, 100(44), 1-8. https://doi.org/10.1101/2021.05.01.21256442

- Sirigatti, S., Stefanile, C., & Menoni, E. (1988). Per un adattamento italiano del Maslach Burnout Inventory (MBI). Bollettino di Psicologia Applicata, 187-188, 33-39. Retrieved from: https://www.researchgate.net/profile/Cristina_Stefanile/publication/233992983_Per_un_adat tamento_italiano_del_Maslach_Burnout_Inventory_MBI_Toward_an_Italian_version_of_th e_Maslach_Burnout_Inventory_MBI/links/0912f510c22cab3bf1000000.pdf
- Soncini, A., Floris, F. & Matteucci, M. C. (2022). Feeling supported and engaged during covid-19: the role of family and colleagues in promoting teachers' well-being. *Teachers and Teaching: Theory and Practice*. <u>https://doi.org/10.1080/13540602.2022.2144820</u>
- Soncini, A., Politi, E. & Matteucci, M. C. (2021). Teachers navigating distance learning during COVID-19 without feeling emotionally exhausted: the protective role of self-efficacy. *School Psychology*, *36*, 494-503. <u>https://doi.org/10.1037/spq0000469</u>
- Topp, C. W., Østergaard, S. D., Søndergaard, S., & Bech, P. (2015). The WHO-5 Well-Being Index: a systematic review of the literature. *Psychotherapy and Psychosomatics*, 84(3), 167-176. <u>https://doi.org/10.1159/000376585</u>
- UNESCO (2020). Teacher Task Force calls to support 63 million teachers touched by the COVID-19 crisis. Retrieved from (last access, 14/07/2022): <u>https://en.unesco.org/news/teacher-task-force-calls-support-63-million-teachers-touched-covid-19-crisis</u>
- UNESCO (2021). UNESCO figures show two thirds of an academic year lost on average worldwide due to Covid-19 school closures. Retrieved from (last access, 14/07/2022): <u>https://en.unesco.org/news/unesco-figures-show-two-thirds-academic-year-lost-average-</u> worldwide-due-covid-19-school
- Toto, G. A., & Limone, P. (2021) Motivation, Stress and Impact of Online Teaching on Italian Teachers during COVID-19. *Computer*, 10, 1-12. https://doi.org/10.3390/ computers10060075

- Truzoli, R., Pirola, V., & Conte, S. (2021). The impact of risk and protective factors on online teaching experience in high school Italian teachers during the COVID-19 pandemic. *Journal* of Computer Assisted Learning, 37(4), 940-952. 2. <u>https://doi.org/10.1111/jcal.12533</u>
- Viac, C., & Fraser, P. (2020). Teachers' well-being: A framework for data collection and analysis, OECD Education Working Papers, 213, OECD Publishing, Paris, https://doi.org/10.1787/c36fc9d3-en.
- York, R. (2018). Control variables and causal inference: A question of balance. *International Journal of Social Research Methodology*, 21(6), 675-684. <u>https://doi.org/10.1080/13645579.2018.1468730</u>
- Zhen, B., Yao, B., & Zhou, X. (2022). Acute stress disorder and job burnout in primary and secondary school teachers during the COVID-19 pandemic: The moderating effect of sense of control. *Current Psychology*, 41, 1-8. <u>https://doi.org/10.1007/s12144-022-03134-7</u>